\documentclass[12pt,twoside]{article}
\renewcommand{\baselinestretch}{1}
\usepackage{amsmath,amsthm,amsfonts,amssymb,latexsym,epsfig}
\usepackage{enumerate}
\usepackage{hyperref}
\setcounter{page}{1}
\setlength{\textheight}{21.6cm}
\setlength{\textwidth}{14cm}
\setlength{\oddsidemargin}{1cm}
\setlength{\evensidemargin}{1cm}
\pagestyle{myheadings}
\thispagestyle{empty}
\markboth{\small{R. Rajendra, P. S. K. Reddy and C. N. Harshavardhana}}
{\small{Stress-Sum Index for Graphs}}
\date{}


\DeclareMathOperator{\T}{\mathcal{T}}
\DeclareMathOperator{\str}{\text{str}}
\DeclareMathOperator{\s}{\mathcal{S}}
\begin{document}
\begin{thebibliography}{99}
	
	\bibitem{bdr} K. Bhargava, N.N. Dattatreya, and R. Rajendra, On stress of a vertex in a graph, \emph{Preprint}, 2020.
	
	\bibitem{gut1} I. Gutman, N. Trinajstić, Graph theory and molecular orbitals. Total $n$-electron energy of alternant hydrocarbons, \emph{Chem. Phys. Lett.}, 17(4) (1972), 535-538.
	
	\bibitem{gut2} I. Gutman, B. Ruščić, N. Trinajstić, C. F. Wilcox, Graph theory and molecular orbitals. XII. Acyclic polyenes, \emph{J. Chem. Phys.}, 62 (1975), 3399-3405.
	
	\bibitem{gut4} I. Gutman, K. C. Das, The first Zagreb index 30 years after, \emph{MATCH Commun. Math. Comput. Chem.}, 50 (2004), 83-92.
	
	\bibitem{hara} F. Harary, \emph{Graph Theory}, Addison Wesley, Reading, Mass, 1972.
	\bibitem{indu2} M. Indhumathy, S. Arumugam, Veeky Baths and Tarkeshwar Singh, Graph theoretic concepts in the study of biological networks, Applied Analysis in Biological and Physical Sciences, Springer Proceedings in Mathematics \& Statistics, 186 (2016), 	187-200.
	\bibitem{raj1} R. Rajendra, P. S. K. Reddy and I. N. Cangul, Stress Index for Graphs, Communicated for Publication 
	\bibitem{Shannon}	P. Shannon, A. Markiel,  O. Ozier,  N.S. Baliga, J.T. Wang, D. Ramage,  N. Amin, 	B. Schwikowski,  T. Idekar, Cytoscape: a software environment for integrated models of biomolecular interaction networks, \emph{Genome Res.}, 13(11) (2003), 2498–2504 
		\bibitem{Shimbel} A. Shimbel,	Structural Parameters of Communication
	Networks, \emph{Bull. Math. Biol.}, 15 (1953), 501-507
	
	\bibitem{Zhou}B. Zhou, Zagreb indices, \emph{MATCH Commun. Math. Comput. Chem.}, 52 (2004), 113–118. 
	
\end{thebibliography}

	
	\footnotetext[1]{$^\ddagger$Corresponding author: pskreddy@jssstuniv.in}
	\centerline {\Large{\bf  Stress-Sum Index for Graphs}}
	
	\centerline{}
	\centerline{\bf {R. Rajendra, P. Siva Kota Reddy and C. N. Harshavardhana}}
	\centerline{}
	\centerline{$^1$Department of Mathematics}
	\centerline{Mangalore University}
	\centerline{Mangalagangothri, Mangalore-574 199, INDIA}
	\centerline{\it Email: rrajendrar@gmail.com}
	\centerline{}
	%\centerline{\bf {}}
	%\centerline{}
	\centerline{$^2$Department of Mathematics}
	\centerline{Sri Jayachamarajendra College of Engineering}
	\centerline{JSS Science and Technology University}
	\centerline{Mysuru-570 006, INDIA}
	\centerline{\it Email: pskreddy@jssstuniv.in}
	\centerline{}
	
	\centerline{$^3$Department of Mathematics}
	\centerline{Government First Grade College for Women}
	\centerline{Holenarasipur-573 211, INDIA}
	\centerline{\it Email: cnhmaths@gmail.com}
	\centerline{}

\theoremstyle{definition}
\newtheorem{Theorem}{Theorem}[section]
\newtheorem{Proposition}[Theorem]{Proposition}
\newtheorem{Definition}[Theorem]{Definition}
\newtheorem{Corollary}[Theorem]{Corollary}
\newtheorem{Lemma}[Theorem]{Lemma}
\newtheorem{Example}[Theorem]{Example}
\newtheorem{Problem}[Theorem]{Problem}
\newtheorem{Remark}[Theorem]{Remark}




\begin{abstract}
	The stress of a vertex is a  node centrality index, which has been introduced by Shimbel (1953). The stress of  a vertex in a graph is the number of geodesics (shortest paths) passing through it. A topological index of a chemical structure (graph) is a number that correlates the chemical structure with chemical reactivity or physical properties. In this paper, we introduce a new topological index for graphs called stress-sum index using stresses of vertices. Further, we  establish some inequalities, prove some results and compute stress-sum index for some standard graphs.
\end{abstract}

\noindent {\bf 2010 Mathematics Subject Classification:} 05Cxx \\
{\bf Keywords:} Graph, neighborhood of a vertex, stress of a vertex, path, geodesic, stress, topological index.

\section{Introduction} \hskip .63cm For standard terminology and notion in graph theory, we follow the text-book of
Harary~\cite{hara}. The non-standard will
be given in this paper as and when required.\\

Let $G=(V,E)$ be a graph (finite and undirected). The distance between two vertices $u$ and $v$ in $G$, denoted by $d(u,v)$ is the number of edges in a shortest path (also called a graph geodesic) connecting them. We say that a graph geodesic $P$ is passing through a vertex $v$ in $G$ if $v$  is an internal vertex of $P$ (i.e., $v$ is a vertex in $P$, but not an end vertex of $P$). For two vertices $u$ and $v$ in $G$, $g(u,v)$ denotes the number of geodesics whose end vertices are $u$ and $v$.\\ 

	The concept of stress of a node (vertex) in a network (graph) has been introduced by Shimbel as centrality measure in 1953 \cite{Shimbel}. This centrality measure has applications in biology, sociology, psychology,  etc., (See \cite{Shannon, indu2}). The stress of  a vertex $v$ in a graph $G$, denoted by $\text{str}_G(v)$ $\text{str}(v)$, is the number of geodesics passing through it. We denote the maximum stress among all the vertices of $G$ by $\Theta_G$ and minimum stress among all the vertices of $G$ by $\theta_G$. Further, the concepts of stress number of a graph and stress regular graphs have been studied by K. Bhargava, N.N. Dattatreya, and R. Rajendra in  their paper \cite{bdr}.
	 The stress number of $G$, denoted by  $N_\text{str}(G)$, is defined as: 
	$$ N_{\str} (G) = \sum_{v\in V} \str(v).$$
	A graph $G$ is $k$-stress regular if $\str(v)=k$ for all $v\in V(G)$.\\
	
	The Zagreb indices have been defined using degrees of vertices in a graph to explain some properties of chemical compounds at molecular level \cite{gut1,gut2}.
	The first Zagreb index $M_1(G)$ 
	and the second Zagreb index $M_2(G)$  
	of a simple graph $G$ are defined as:
	\begin{align} \label{M1}
	M_1(G) &= \sum_{v\in V(G)} {d(v)^2}\\
\label{M2}
	M_2(G) &= \sum_{uv\in E(G)} {d(u)d(v)}.
	\end{align}
	
		
	By the motivation of these indices, Rajendra et al. \cite{raj1}  have introduced two  topological indices of for graphs called  first  stress index and second  stress index,  using stresses of vertices. The first stress index $\s_1(G)$ 
	and the second stress index $\s_2(G)$  
	of a simple graph $G$ are defined as
	\begin{align} \label{S1}
	\s_1(G) &= \sum_{v\in V(G)} {\str(v)^2}\\
	\label{S2}
	\s_2(G) &= \sum_{uv\in E(G)} {\str(u)\str(v)}.
	\end{align}

We note that the  first Zagreb index $M_1(G)$  satisfies the identity 	
		\begin{align}\label{M3}
		M_1(G) = \sum_{uv\in E(G)} {d(u)+d(v)}
		\end{align}
but $\s_1(G)$ does not satisfy such identity. For instance, consider the path $P_3$ on $3$ vertices.

\begin{figure}[h!]
%	\vskip 1.3cm 
\unitlength 1mm % = 2.845pt
\linethickness{0.4pt}
\ifx\plotpoint\undefined\newsavebox{\plotpoint}\fi % GNUPLOT compatibility
\begin{picture}(20,20)(-49,135)
\thicklines
\put(0,150){\circle*{2}}
\put(20,150){\circle*{2}}
\put(40,150){\circle*{2}}
\put(0,150){\line(1,0){40}}
\put(0,153){\makebox(0,0)[cc]{$v_1$}}
\put(20,153){\makebox(0,0)[cc]{$v_2$}}
\put(40,153){\makebox(0,0)[cc]{$v_3$}}
\end{picture}
\vskip -1cm
\caption{The path $P_3$.}
\label{fig-2}
\end{figure} 

The stresses of the vertices of $P_3$ are as follows: $\text{str}(v_1) = \text{str}(v_3) = 0$ and $\text{str}(v_2) =1$. The first stress index of $P_3$ is, $$\s_1(P_3) = \str(v_1)^2+\str(v_2)^2 +\str(v_3)^2 = 0^2 + 1^2 + 0^2 =1.$$
But 
$$\sum_{uv\in E(P_3)} {\str(u)+\str(v)} = \str(v_1)+\str(v_2) + \str(v_2) + \str(v_3) = 0+1+1+0 =2.$$
Therefore there is a scope for introducing a new topological index using stress on vertices which is motivated by the identity~\eqref{M3}. In this paper we introduce such  topological index for graphs using stress on vertices called stress-sum index. Further, we  establish some inequalities and compute  stress-sum index for some standard graphs.
		
\section{Stress-Sum Index for Graphs}
\allowdisplaybreaks
\begin{Definition} \label{S-Sindex}
	The stress-sum index $\s\s(G)$ 
	of a simple graph $G$ is defined as
\begin{align}\label{ss}
\s\s(G) &=  \sum_{uv\in E(G)} {\str(u)+\str(v)}
\end{align}
\end{Definition}

\noindent {\bf Observation}: From the Definition~\ref{S-Sindex}, it follows that, for any graph $G$,
$$2m \theta_G\leq  \s\s(G) \leq 2m \Theta_G$$
where $m$ is the number of edges in $G$.

\begin{Example}\label{eg3}
	Consider the graph $G$ given in Figure~\ref{fig-1}. 
		\begin{figure}[h!]
			\vskip -1cm
		\unitlength 1mm % = 2.845pt
		\linethickness{0.4pt}
		\ifx\plotpoint\undefined\newsavebox{\plotpoint}\fi % GNUPLOT compatibility
		\begin{picture}(140,34)(-40,183)
		%%%%%%%%%%PG(Z8)%%%%%%%%%%%%%%%%
		\put(7,200){\circle*{2}}
		\put(28,200){\circle*{2}}
		\put(49,200){\circle*{2}}
		
		\put(7,180){\circle*{2}}
		
		\put(28,180){\circle*{2}}
		\put(49,180){\circle*{2}}
		\put(7,192){\circle*{2}}
		\put(49,191.5){\circle*{2}}
		
		\put(7,200){\line(1,0){20}}
		\put(28,200){\line(1,0){20}}
		%\put(7,180){\line(1,-1){20}}
		%\put(28,160){\line(1,1){20}}
		
		\put(28,180){\line(0,1){20}}
		\put(7,180){\line(1,1){20}}
		\put(7,180){\line(1,0){20}}
		\put(28,180){\line(1,0){20}}
		\put(49,179.5){\line(-1,1){20}}
		
		\multiput(7,192)(.5,.20){43}{\line(1,0){0.5}}
		\multiput(49,191.5)(-0.5,.20){43}{\line(1,0){0.5}}
		
		
		\put(7,204){\makebox(0,0)[cc]{$v_1$}}
		\put(28,204){\makebox(0,0)[cc]{$v_2$}}
		\put(49,204){\makebox(0,0)[cc]{$v_3$}}
		
		\put(7,176){\makebox(0,0)[cc]{$v_4$}}
		\put(28,176){\makebox(0,0)[cc]{$v_5$}}
		\put(49,176){\makebox(0,0)[cc]{$v_6$}}
		\put(7,189){\makebox(0,0)[cc]{$v_7$}}
		\put(49,189){\makebox(0,0)[cc]{$v_8$}}
		\put(28,170){\makebox(0,0)[cc]{$G$}}
		\end{picture}
		\vskip 1.3cm 
		\caption{A graph $G$}
		\label{fig-1}
	\end{figure}
	
The stresses of the vertices of $G$ are as follows:\\
	$\text{str}(v_1) = \text{str}(v_3) = \text{str}(v_7) = \text{str}(v_8) =0$,\\
		$\text{str}(v_2) = 19$,\\
		$\text{str}(v_5) =1$,\\
		$\text{str}(v_4)  = \text{str}(v_6) = 0$. 
		
The stress-sum index of $G$ is: 
	\begin{align*} 
	\s\s(G) =& (\str(v_2)+\str(v_1))+(\str(v_2)+\str(v_3))+(\str(v_2)+\str(v_7))\\
	&+(\str(v_2)+\str(v_8)+(\str(v_2)+\str(v_4))+(\str(v_2)+\str(v_5))\\
	&+(\str(v_2)+\str(v_6))+(\str(v_4)+\str(v_5))+(\str(v_5)+\str(v_6))\\
	=&(19+0) + (19 + 0) +  (19 + 0) +  (19 + 0)  + (19 + 0) + (19 + 1) \\
	&+ (19 + 0)+ (0 + 1) + (1 + 0)\\
	 =& 136.
	\end{align*}
\end{Example}


\begin{Proposition}\label{Prop1}
	Let $N$ be the number of geodesics of length  $\geq2$ in a graph $G$. Then
	\begin{align}	\label{ineq1.2} 
	0 \leq \s\s(G) \leq 2N (|E|-t),
	\end{align}	
where $t$ is the number of edges with end vertices having zero stress  in $G$.
\end{Proposition}
\begin{proof}
	If $N$ is the number of all geodesics of length  $\geq2$ in a graph $G$, then by the definition of stress of a vertex, for any vertex $v$ in $G$, $ 0 \leq \text{str}(v) \leq N$. Hence by the Definition~\ref{S-Sindex}, we have 	
	\begin{align} \label{ineq2}
	0 \leq \s\s(G) \leq 2N (|E|-t),
	\end{align}	
where $t$ is the number of edges with end vertices having zero stress  in $G$.
	\end{proof}

\begin{Corollary}\label{Cor1}
	If there is no geodesic of length  $\geq2$ in a graph $G$, then $\s\s(G) = 0 $. Moreover, for a complete graph $K_n$, $\s\s(K_n)=0$.
\end{Corollary}
\begin{proof}
	If there is no geodesic of length  $\geq2$ in a graph $G$, then $N =0$. Hence,  by the Proposition~\ref{Prop1}, we have $\s\s(G) = 0 $.
	
	In $K_n$,  there is no geodesic of length  $\geq2$ and so $\s\s(K_n)=0$.
\end{proof}

\begin{Theorem}\label{thm-1}
	For a graph $G$,  $\s\s(G) = 0$ if and only if neighbours of every vertex induce a complete subgraph of $G$.
\end{Theorem}

\begin{proof}
	Suppose that $\s\s(G) = 0$. Then by the Definition~\ref{S-Sindex}(Eq.\eqref{S1}), \linebreak $\str(u)+\str(v) =0$, $\forall uv\in E(G)$. Hence  $\str(v) =0$,  $\forall v\in V(G)$. Let $ v\in V(G)$. We need to show that neighbors of $v$ induce a complete subgraph of $G$. If $v$ is a pendant vertex, then there is nothing to prove. Suppose that $v$ is not a pendant vertex. We claim that any two neighbouring vertices are adjacent in $G$. If there are two neighbours $u$ and $w$ of $v$ that are not adjacent in $G$, then $uvw$ is a graph geodesic passing through $v$, which implies $\str(v) \geq 1$, a contradiction. 	Hence our claim holds. Thus neighbours of $v$ induce a complete subgraph of $G$. Since $v$ is arbitrary in $V(G)$, the neighbours of every vertex induce a complete subgraph of $G$.
	
	Conversely, suppose that neighbours of every vertex in $G$ induce a complete subgraph of $G$. Let $v \in V(G)$. Since neighbors of $v$ induce a complete subgraph of $G$, any two neighbouring vertices are adjacent and so there is no geodesic of length $\geq 2$ passing through $v$. Since  $v$ is an arbitrary vertex in $G$,  by the Corollary~\ref{Cor1}, it follows that  $\s\s(G) = 0$.	
\end{proof}


\begin{Proposition}\label{Prop-2}
	For the complete bipartite $K_{m,n}$,
	$$\s\s(K_{m,n})=\dfrac{mn}{2}\left[{n(n-1)}+{m(m-1)}\right].$$
\end{Proposition}
\begin{proof}
	Let	$V_1 =\{v_1,\ldots, v_m\}$ and $V_2 =\{u_1,\ldots, u_n\}$ be the partite sets of $K_{m,n}$. We have,
	\begin{align}\label{v1}
	\str(v_i) = \dfrac{n(n-1)}{2}~ \text{for $1\leq i \leq m$}
	\end{align}
	and \begin{align}\label{v2}
	\str(u_j) = \dfrac{m(m-1)}{2}~ \text{for $1\leq j \leq n$.}
	\end{align}
	Using \eqref{v1} and \eqref{v2} in the Definition~\ref{S-Sindex}, we have
		\begin{align*} 
	\s\s(K_{m,n}) &= \sum_{uv\in E(G)} {\str(u) + \str(v)}\\
	& = \sum_{1\leq i \leq m, ~1\leq j \leq m} {\str(v_i)}+{\str(u_j)} \\
	& =\sum_{1\leq i \leq m, ~1\leq j \leq n}\left[\dfrac{n(n-1)}{2}+\dfrac{m(m-1)}{2}\right] \\
	& =mn\left[\dfrac{n(n-1)}{2}+\dfrac{m(m-1)}{2}\right]\\
	& =\dfrac{mn}{2}\left[{n(n-1)}+{m(m-1)}\right].
	\end{align*}
	\vskip -1cm
\end{proof}

\begin{Proposition}\label{Prop-3}
	If $G=(V,E)$ is a $k$-stress regular graph, then 
	$$\s\s(G)  = 2k|E|.$$
\end{Proposition}
\begin{proof}
	Suppose that $G$ is a $k$-stress regular graph. Then\\ 
	\centerline{$\str(v)=k$ for all $v\in V(G)$.}\\
	By the Definition~\ref{S-Sindex}, we have
	\begin{align*} 
	\s\s(G) &= \sum_{uv\in E(G)} {\str(u)+\str(v)}\\
	& =\sum_{uv\in E(G)} {k+ k}\\
	& =2k |E|.
	\end{align*}
	\vskip -1cm
\end{proof}
~
\begin{Corollary}\label{Cor-3A}
	For a cycle $C_{n}$,
	$$\s\s(C_{n}) = \begin{cases}
	\dfrac{n(n-1)(n-3)}{4}, & \text{if $n$ is odd}\\[2ex]
	\dfrac{n^2(n-2)}{4}, & \text{if $n$ is even}.\end{cases} 	 $$
\end{Corollary}

\begin{proof}
	For any vertex $v$ in $C_n$, we have,   
	$$\str(v) = \begin{cases}
	\dfrac{(n-1)(n-3)}{8}, & \text{if $n$ is odd}\\[2ex]
	\dfrac{n(n-2)}{8}, & \text{if $n$ is even}.\end{cases} $$
	Hence  $C_n$ is 
	$$\begin{cases}
	\dfrac{(n-1)(n-3)}{8}\text{-stress regular}, & \text{if $n$ is odd}\\[2ex]
	\dfrac{n(n-2)}{8}\text{-stress regular}, & \text{if $n$ is even}.\end{cases} $$
	Since $C_n$ has $n$ vertices and $n$ edges,  by the Proposition~\ref{Prop-3}, we have
	\begin{align*} 	\s\s(C_{n})  &= 2n \times \begin{cases}
	\dfrac{(n-1)(n-3)}{8}, & \text{if $n$ is odd}\\[2ex]
	\dfrac{n(n-2)}{8}, & \text{if $n$ is even}.\end{cases} \\
	&= \begin{cases}
	\dfrac{n(n-1)(n-3)}{4}, & \text{if $n$ is odd}\\[2ex]
	\dfrac{n^2(n-2)}{4}, & \text{if $n$ is even}.\end{cases}
	\end{align*}	
	\vskip -1cm
\end{proof}


\begin{Proposition} \label{prop2-12}
	Let $T$  be a tree on $n$ vertices. Then 
		\begin{align*}	\s\s(T) =&	\sum_{uv\in J} \left[\sum_{1\leq i<j\leq m(u)} |C_i^u||C_j^u| + \sum_{1\leq i<j\leq m(v)} |C_i^v||C_j^v| \right]\\
	&~~~~~~~~~~~~~ +  \sum_{w \in Q} \sum_{1\leq i<j\leq m(w)} |C_i^w||C_j^w|.
	\end{align*}
	where $J$ is the set of internal(non-pendant) edges in $T$, $Q$ denotes the set of all vertices adjacent to pendent vertices in $T$, and the sets $C_{1}^v,\dots, C_m^v$ denotes the vertex sets of the components of $T-v$ for an internal vertex $v$ of degree $m=m(v)$.
\end{Proposition}
\begin{proof}
	We know that a pendant vertex in $T$ has zero stress. 
	Let $v$ be an internal vertex of $T$ of degree $m=m(v)$. Let $C_{1}^v,\dots, C_m^v$ be the components of $T-v$.  Since there is only one path between any two vertices in a tree, it follows that, 
	\begin{align}\label{Qstress}
	\text{str}(v)=\sum_{1\leq i<j\leq m} |C_i^v||C_j^v|
	\end{align}
Let	$J$ denotes the set of internal(non-pendant) edges, and $P$ denotes pendant edges and $Q$ denotes the set of all vertices adjacent to pendent vertices in $T$. Then using \eqref{Qstress} in the Definition~\ref{S-Sindex} (\eqref{ss}), we have
	\begin{align*}	\s\s(T) =& \sum_{uv\in J} {\str(u)+\str(v)} + \sum_{uv\in P} {\str(u)+\str(v)}\\
		=& \sum_{uv\in J} {\str(u)+\str(v)} + \sum_{w \in Q} {\str(w)}\\
		=&	\sum_{uv\in J} \left[\sum_{1\leq i<j\leq m(u)} |C_i^u||C_j^u| + \sum_{1\leq i<j\leq m(v)} |C_i^v||C_j^v| \right]\\
		&~~~~~~~~~~~~~  +  \sum_{w \in Q} \sum_{1\leq i<j\leq m(w)} |C_i^w||C_j^w|.
		\end{align*}
			\vskip -1cm
\end{proof}
~
\begin{Corollary}\label{Cor2-12}
	For the path $P_n$ on $n$ vertices
	$$\s\s(P_n) =\frac{1}{3}n(n-1)(n-2).$$
\end{Corollary}
\begin{proof} The proof of this corollary follows by above Proposition~\ref{prop2-12}. We follow the proof of the Proposition~\ref{prop2-12} to compute the index. Let $P_n$ be the path with vertex sequence $v_1, v_2,\ldots, v_n$ (shown in  Figure~\ref{fig-3}).
	\begin{figure}[h!]
		%	\vskip 1.3cm 
		\unitlength 1mm % = 2.845pt
		\linethickness{0.4pt}
		\ifx\plotpoint\undefined\newsavebox{\plotpoint}\fi % GNUPLOT compatibility
		\begin{picture}(20,20)(-20,135)
		\thicklines
		\put(0,150){\circle*{2}}
		\put(20,150){\circle*{2}}
		\put(40,150){\circle*{2}}
		\put(60,150){\circle*{2}}
		\put(80,150){\circle*{2}}
		\put(100,150){\circle*{2}}
		
		\put(0,150){\line(1,0){60}}
		%\put(60,150){\line(1,0){20}}
		\put(70,150){\makebox(0,0)[cc]{$.~. ~ . ~.~.$}}
		\put(80,150){\line(1,0){20}}
		
		\put(0,153){\makebox(0,0)[cc]{$v_1$}}
		\put(20,153){\makebox(0,0)[cc]{$v_2$}}
		\put(40,153){\makebox(0,0)[cc]{$v_3$}}
		\put(60,153){\makebox(0,0)[cc]{$v_4$}}
		\put(80,153){\makebox(0,0)[cc]{$v_{n-1}$}}
		\put(100,153){\makebox(0,0)[cc]{$v_n$}}
		\put(54,140){\makebox(0,0)[cc]{$P_n$}}
		%\qbezier(31.75,169.75)(61.5,173.875)(40.25,159.5)
		\end{picture}
		\vskip -0.3cm
		\caption{The path $P_n$ on $n$ vertices.}
		\label{fig-3}
	\end{figure}
	
	We have,	
	$$\text{str}(v_i)=(i-1)(n-i),~~~~~~1\leq i \leq n.$$
	Then 
	\begin{align*}
	\s\s(P_n) &= \sum_{uv\in E(P_n)} {\str(u)+\str(v)}\\
	&= \sum_{i=1}^{n-1} {\str(v_{i})+\str(v_{i+1})}\\
	& = \sum_{i=1}^{n-1} [(i-1)(n-i) +(i)(n-i-1)]\\
	&=\frac{1}{3}n(n-1)(n-2).
	\end{align*}
	\vskip -1cm	
\end{proof}
~

\begin{Proposition} \label{prop2.14}
	Let $Wd(n,m)$ denotes the windmill graph constructed for $n \geq 2$ and $m \geq 2$ by joining $m$ copies of the complete graph $K_n$ at a shared universal vertex $v$. 
	Then
	$$\s\s(Wd(n,m)) = \frac{n^3 m(m-1)}{2}.$$
	Hence, for the friendship  graph $F_k$ on $2k+1$ vertices,
	$$\s\s(F_k) = 4k(k-1).$$
\end{Proposition}
\begin{proof}
	Clearly the stress of any vertex other than universal vertex is zero in $Wd(n,m)$, because neighbors of that vertex induces a complete subgraph of $Wd(n,m)$. Also, since there are $m$ copies of $K_n$ in $Wd(n,m)$ and their vertices are adjacent to $v$, it follows that, the only geodesics passing through $v$ are of length $2$ only. So, $\str(v)=n^2(m-1)/2$. 
	Note that there are $mn$ edges incident on $v$ and the edges that are not incident on $v$ have end vertices of stress zero. Hence by the Definition~\ref{S-Sindex}, we have
		\begin{align*}
\s\s(Wd(n,m)) &= mn \str(v)\\
& =mn[n^2(m-1)/2]\\
& = \frac{n^3 m(m-1)}{2}.
	\end{align*}
Since the friendship  graph $F_k$ on $2k+1$ vertices is nothing but $Wd(2,k)$, it follows that 
	$$\s\s(F_k) = \frac{2^3 k(k-1)}{2} = 4k(k-1).$$
		\vskip -0.8cm	
\end{proof}
~
\begin{Proposition} \label{prop2.15}
	Let $W_n$ denotes the wheel graph constructed on $n \geq 4$ vertices.
	Then
	$$\s\s(W_n) = \begin{cases}
	\dfrac{\left(n-1\right)\left(7n-10\right)\left(n-4\right)}{8}, & \text{if $n$ is even;}\\[2ex]
	\dfrac{\left(n-1\right)^2\left(7n-25\right)}{8}, & \text{if $n$ is odd}.\end{cases}  .$$
\end{Proposition}
\begin{proof}
	In $W_n$ with $n \geq 4$, there are $(n-1)$ peripheral vertices and one central vertex, say $v$. It is easy to see that 
	\begin{equation}\label{cntr}
	\str(v) = \frac{(n-1)(n-4)}{2}
	\end{equation}
	Let $p$ be a peripheral vertex. Since $v$ is adjacent to all the peripheral vertices in $W_n$, there is no geodesic passing through $p$ and containing $v$. Hence contributing vertices for $\str(p)$ are the rest peripheral vertices. So, by denoting the cycle $W_n -p$ (on $n-1$ vertices)  by $C_{n-1}$, we have 
	\begin{align}
	\str_{W_n}(p) &= \str_{W_n -v}(p)\nonumber\\
	& =  \str_{C_{n-1}}(p) \nonumber\\
	&= \begin{cases}
	\dfrac{(n-2)(n-4)}{8}, & \text{if $n-1$ is odd; }\\[2ex]
	\dfrac{(n-1)(n-3)}{8}, & \text{if $n-1$ is even,}\end{cases}  \nonumber\\
	&= \begin{cases}
	\dfrac{(n-2)(n-4)}{8}, & \text{if $n$ is even;}\\[2ex]
	\dfrac{(n-1)(n-3)}{8}, & \text{if $n$ is odd}.\end{cases}  \label{peri}
	\end{align}
	Let us denote the set of all radial edges in $W_n$ by $R$, and the set of all peripheral edges by $Q$. Note that there are $(n-1)$ radial edges and $(n-1)$ peripheral edges in $W_n$.
	Using \eqref{cntr} and \eqref{peri} in the Definition~\ref{S-Sindex}, we have
	\begin{align*} 
	\s\s(W_n) =& \sum_{xy\in R} [{\str(x)+\str(y)}]+ \sum_{xy\in Q} [{\str(x)+\str(y)}]\\
	=& (n-1) [{\str(v)+\str(p)}] + (n-1)\cdot 2 \cdot  {\str(p)} \\
	=& (n-1) \left[{\frac{(n-1)(n-4)}{2}+\begin{cases}
		\dfrac{(n-2)(n-4)}{8}, & \text{if $n$ is even;}\\[2ex]
		\dfrac{(n-1)(n-3)}{8}, & \text{if $n$ is odd}.\end{cases}}\right] \\
	& ~~+ 2(n-1) \times \begin{cases}
	\dfrac{(n-2)(n-4)}{8}, & \text{if $n$ is even;}\\[2ex]
	\dfrac{(n-1)(n-3)}{8}, & \text{if $n$ is odd}.\end{cases}  \\
	=&  \begin{cases}
	\dfrac{(n-1)^2(n-4)}{2}+\dfrac{3(n-1) (n-2)(n-4)}{8}, & \text{if $n$ is even;}\\[2ex]
	\dfrac{(n-1)^2(n-4)}{2}+\dfrac{3(n-1)^2(n-3)}{8}, & \text{if $n$ is odd}.\end{cases} \\
	=&  \begin{cases}
		\dfrac{\left(n-1\right)\left(7n-10\right)\left(n-4\right)}{8}, & \text{if $n$ is even;}\\[2ex]
	\dfrac{\left(n-1\right)^2\left(7n-25\right)}{8}, & \text{if $n$ is odd}.\end{cases} 
		\end{align*}
	\vskip -0.9cm	
\end{proof}
\subsection*{Conclusion} We have introduced a new topological index for graphs called stress-sum index using stresses of vertices. Further, we  established some inequalities, proved some results and computed  the stress-difference index for some standard graphs. A large number of molecular-graph-based structure descriptors (topological indices) have been defined, depending on vertex degrees. But in this paper, we have defined the new topological index for graphs without using the degrees of vertices. This index can be used to determine $\s\s(G)$ for other classes of graphs and results in this direction will be reported in a subsequent paper. 
\pagebreak


\end{document}

